
Daily status, June 7
SOAS ISFS Status, June 7, 2013

32m Licor is still down.  We will remove it from the tower for trouble-shooting in the base.  Options are: (1) receive loaner from Licor, (2) replace with 
20m Licor and add krypton hygrometer at 20m.  There is a spare sonic with a serializer (SN 0537) in the base.

Rain last night and foggy this morning.  More rain likely today.  Everything very wet.

Sonics: 13.9m and 32m sonics still wet and not working yet.

TRH, P, p: TRH at pond not reporting; P okay.

Wind speed & direction: See sonic report; pond U,V okay.

Soil, Rainr: tb.pond not reporting

        Vmote.(rad,a,b) ~ 5.9, 9.1, 11.9 V.

Radiation, Wetness: Tsfc.32m not seen on cockpit, but okay.  

                    Tsfc.8m intemittent over past 24 hours.

h2o, co2: 32m Licor not reporting.  kryptons still wet.

System (GPS, Vdsm): ok 
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